Author's response to reviews

Title: Reliability of upright posture measurements in primary school children

Authors:

Maureen P McEvoy (maureen.mcevoy@unisa.edu.au)
Karen Grimmer (karen.grimmer@unisa.edu.au)

Version: 6 Date: 6 June 2005

Author's response to reviews:

RESPONSE TO REVIEWERS’ REPORTS
Due 6th June 2005
Title: Reliability of upright posture measurements in primary school children

Authors: Maureen McEvoy and Karen Grimmer

Reviewer's Comment:
The reviewer has requested that standard error (SE) be replaced with standard deviation (SD) as this is the appropriate measure of variability in these types of studies. The reviewer has trusted the Authors to make the changes and he does not need to review the revisions.

Authors’ Response:
Table 3 has been modified and SD has been included and SE has been removed. Reference to SE in the text has been replaced by SD.

We would like to thank the reviewers for their support and patience throughout the review process for this article.

Maureen McEvoy and Karen Grimmer